

# Documents de Treball

MARKET REWARDS TO PATTERNS ON INCREASING EARNINGS: DO CASH FLOW PATTERNS, ACCRUALS MANIPULATION AND REAL ACTIVITIES MANIPULATION MATTER?

Su-Ping Liu, Juan Manuel García Lara

Document de Treball núm.13/3

Departament d'Economia de l'Empresa

© Su-Ping Liu, Juan Manuel García Lara Coordinador / Coordinator *Documents de treball:* 

David Urbano

http://selene.uab.es/dep-economia-empresa/dt

e-mail: <u>david.urbano@uab.es</u> Telèfon / Phone: +34 93 5814298

Fax: +34 93 5812555

# Edita / Publisher:

Departament d'Economia de l'Empresa http://selene.uab.es/dep-economia-empresa/ Universitat Autònoma de Barcelona Facultat de Ciències Econòmiques i Empresarials Edifici B 08193 Bellaterra (Cerdanyola del Vallès), Spain Tel. 93 5811209 Fax 93 5812555

#### ISSN:

1988-7736. Documents de Treball (Departament d'Economia de l'Empresa, Universitat Autònoma de Barcelona)

# MARKET REWARDS TO PATTERNS ON INCREASING EARNINGS: DO CASH FLOW PATTERNS, ACCRUALS MANIPULATION AND REAL ACTIVITIES MANIPULATION MATTER?

Su-Ping Liu, Juan Manuel García Lara

Document de Treball núm.13/3

La sèrie *Documents de treball d'economia de l'empresa* presenta els avanços i resultats d'investigacions en curs que han estat presentades i discutides en aquest departament; això no obstant, les opinions són responsabilitat dels autors. El document no pot ser reproduït total ni parcialment sense el consentiment de l'autor/a o autors/res. Dirigir els comentaris i suggerències directament a l'autor/a o autors/res, a la direcció que apareix a la pàgina següent.

A Working Paper in the *Documents de treball d'economia de l'empresa* series is intended as a mean whereby a faculty researcher's thoughts and findings may be communicated to interested readers for their comments. Nevertheless, the ideas put forwards are responsibility of the author. Accordingly a Working Paper should not be quoted nor the data referred to without the written consent of the author. Please, direct your comments and suggestions to the author, which address shows up in the next page.

# Market Rewards to Patterns of Increasing Earnings: Do Cash Flow Patterns, Accruals Manipulation and Real Activities Manipulation Matter?

Su-Ping Liu Universidad Carlos III de Madrid C/Madrid 126, 28903 Getafe, Madrid, Spain

E-mail: sliu@emp.uc3m.es

Juan Manuel García Lara
Universidad Carlos III de Madrid
C/Madrid 126, 28903 Getafe, Madrid, Spain

E-mail: jmglara@emp.uc3m.es

<sup>\*</sup> We are grateful for the helpful comments and suggestions received from Michael Brennan, Beatriz García Osma, Peter Pope, William Rees, Ole-Kristian Hope, and workshop participants at Universidad Carlos III de Madrid, the 28<sup>th</sup> European Doctoral Colloquium in Accounting, the 35<sup>th</sup> European Accounting Association Annual Congress, the XV Spain Doctoral Colloquium in Accounting, and the VIII International Accounting Research Symposium. We also acknowledge financial assistance from the Spanish Ministry of Education and Science (ECO2010-19314), and from the Government of Comunidad de Madrid (Grant CCG10-UC3M/HUM- 4760).

# Market Rewards to Patterns of Increasing Earnings: Do Cash Flow Patterns, Accruals Manipulation and Real Activities Manipulation Matter?

#### **Abstract**

This study explores whether firms have differential price-earnings multiples associated with their means of achieving a sequential pattern of increasing positive earnings. Our main findings show that market participants assign higher price-earnings multiples to firms when their pattern of increasing earnings is supported by the same pattern of increasing cash flows. Market participants assign lower price-earnings multiples to firms suspect of having engaged in accrual-based earnings management, sales manipulation, and overproduction to achieve the earnings pattern. We find, however, that market participants do not penalize firms suspect of having achieved the earnings pattern through the opportunistic reduction of discretionary expenses.

#### 1.Introduction

We empirically examine whether the market prices differently firms with a pattern of increasing earnings that at the same time (i) report the same pattern of increasing cash flows, (ii) have discretionary accruals that if not included in earnings would break the earnings trend, and (iii) alter optimal operational transactions to avoid breaking the earnings pattern. Prior studies show that market participants reward firms that meet or beat certain earnings benchmarks: prior year earnings (e.g., Barth et al., 1999; Francis et al., 2003; Myers et al., 2007; Koonce and Lipe, 2010), and analysts' forecasts (e.g., Kasznik and McNichols, 2002; Bartov et al., 2002). The results of prior research on whether the market rewards differently firms that manage their accounting numbers to meet or beat these targets are mixed. There is evidence showing that price-earnings multiples are reduced when a consecutive string of increasing earnings is supported by earnings management (Francis et al., 2003), and that abnormal returns do not exist for firms that meet or beat earnings forecasts through accruals management and real activities management in the UK (Athanasakou et al., 2011). In contrast, some evidence shows that the market fails to identify firms meeting or beating analysts' forecasts through earnings management (e.g., Bartov et al., 2002; Chen et al., 2010), and that analysts are unable to see through firms that employ earnings management to avoid losses and earnings decreases (Burgstahler and Eames, 2003).

Given these mixed results we contribute to prior literature analyzing whether firms with obvious signals of not having (having) engaged in earnings management to meet or beat the benchmarks receive an additional market reward (penalty). We refer to firms with obvious signals of not having managed earnings as non-suspect beaters. We define non-suspect beaters as those firms with five years of consecutive increases both in earnings and in cash flows. We identify two types of suspect firms: (a) firms with five

years of consecutive earnings increases that present discretionary accruals that, if not included in earnings, would fail to report the pattern of increasing earnings, and (b) firms with five years of consecutive earnings increases that engage in inefficient operational practices with the objective of managing earnings, and that would have broken the pattern of increasing earnings in the absence of those inefficient operational practices. We expect suspect firms to be penalized by the market.

Using a sample of 22,605 US listed non-financial, non-utility, and profit-making firm-year observations for the period 1995-2007, obtained from COMPUSTAT, our findings are consistent with Barth et al. (1999) that firms with a five-year pattern of increasing earnings have higher price-earnings multiples than other firms. Our empirical analysis also generates the following major findings. First, market participants assign higher price-earnings multiples to non-suspect beaters. Second, market participants assign lower price-earnings multiples to suspect firms that use accrual-based earnings management to achieve a five-year pattern of increasing earnings. Third, the price-earnings multiples are reduced when the earnings pattern is achieved through sales manipulation, that is, through increasing sales on credit beyond whatever is advisable by common practice. In addition, market participants assign lower price-earnings multiples to firms suspect of having increased production more than necessary. However, the market fails to reduce rewards to firms suspect of producing the earnings pattern through reductions in discretionary expenses, such as R&D expenses, and selling, general, administrative and advertising expenses.

The remainder of the paper is organized as follows: section 2 provides a discussion of the related literature and describes the hypotheses. Section 3 contains the research design. Section 4 describes the data and the empirical results. Section 5 reports robustness tests. Finally, section 6 summarizes and concludes.

# 2. Prior Research and Development of the Hypotheses

Recent research provides evidence that market participants reward firms that meet or beat certain earnings benchmarks, such as prior year earnings (e.g. Barth et al., 1999; Francis et al., 2003; Myers et al., 2007; Koonce and Lipe, 2010), and analysts' forecasts (e.g., Kasznik and McNichols, 2002; Bartov et al., 2002; Koonce and Lipe, 2010). Regarding earnings increases, prior research shows that benchmark beating firms enjoy higher price-earnings multiples (Barth et al. 1999, Francis et al., 2003) and positive abnormal returns (Myers et al., 2007). Koonce and Lipe (2010) conduct several experiments to investigate how and why investors react to patterns of increasing earnings and conclude that investors assign higher stock price to these firms because they consider a pattern of earnings increases as a signal of higher management's credibility and better future prospects. Overall, this evidence is consistent with the results in the survey study in Graham et al. (2005) that managers seek to meet or beat earnings benchmarks to build credibility and to inform about future growth prospects.

Among the firms that report a pattern of consecutive increases in earnings, some may achieve the pattern through intrinsic performance, while others may engage in different forms of earnings management to achieve the benchmark and mislead investors (Burgstahler and Dichev, 1997; Degeorge et al., 1999). Given that cash flows are more difficult and more costly to manage than accruals, we use them to assess whether the pattern of increasing earnings is genuine. We argue that a firm with a pattern of increasing earnings is non-suspect of having engaged in earnings management whenever the earnings pattern comes together with a pattern of increasing cash flows. If financial analysts use this simple signal to identify non-suspect beaters, we argue that these non-suspect beaters (with patterns of consecutive increases both in earnings and

in cash flows) will enjoy higher price-earnings multiples than firms with only a pattern of increasing earnings. This leads to the first hypothesis in the paper:

**H1:** Investors assign higher price-earnings multiples to firms with a pattern of increasing earnings that is supported by the same pattern of increasing cash flows.

Our second set of hypotheses refers to whether market participants price differently benchmark beating firms suspect of meeting or beating the target through the management of accruals and/or the management of real operational activities. A large number of studies starting with Burgstahler and Dichev (1997), Degeorge et al. (1999) and Kasznik (1999) show that managers engage in earnings management to meet or beat several earnings thresholds, including zero earnings, last year's earnings, and analysts' and managerial earnings forecasts.

Earnings management can be achieved through accounting choices and estimates about accruals, and/or through altering reported earnings by adjusting the timing and the scale of underlying real business activities to mislead market participants. Burgstahler and Dichev (1997) conclude that firms manage working capital accruals to avoid reporting negative earnings, and Kasznik (1999) shows that when managers make optimistic disclosures about future earnings there are significant levels of positive discretionary accruals because managers use accruals opportunistically to achieve their own earnings forecasts. Regarding the manipulation of real operational activities to meet financial reporting goals, Roychowdhury (2006) finds empirical evidence that firms avoid reporting losses through temporal sales increases, overproduction, and the opportunistic reduction of discretionary expenses, such as R&D investment, selling, general and administrative (SGA) expenses and advertising expenses. The survey study conducted by Graham et al. (2005) shows that the majority of managers employ real

activities management to meet or beat earnings benchmarks even though real activities management decreases future cash flows and firm value (Roychowdhury, 2006). Finally, Myers et al. (2007) argue that the frequency of firms reporting patterns of increasing earnings is much larger than it would be expected, and they interpret this preliminary evidence as consistent with earnings management to achieve the patterns. They further show that firms reporting patterns of increasing earnings are more likely to maintain or extend the patterns through various means of earnings management, including reporting more positive or negative special items, increasing stock repurchases and adjusting effective tax rates.

Although there is plenty of evidence showing that firms manage earnings to achieve earnings benchmarks, it is not clear whether market participants penalize benchmark beating firms that meet the benchmarks through earnings management. Francis et al. (2003) show that price-earnings multiples are reduced or are even eliminated when the pattern of meeting last year's earnings is supported by low quality earnings, suggesting that the market is aware of earnings that are managed. Athanasakou et al. (2011) show that the market does not reward UK firms meeting or beating earnings forecasts through accrual-based earnings management and real activities management, concluding that the market provides no incentives for managers to achieve earnings expectations through earnings management. In contrast with this evidence, other studies show that market participants do not penalize firms that meet or beat certain earnings benchmarks through earnings management. Burgstahler and Eames (2003) demonstrate that analysts are unable to consistently see through firms that have engaged in earnings management to avoid earnings decreases, and Bartov et al. (2002) show that the returns to firms that meet or beat analysts' forecasts are only marginally affected depending on whether earnings are managed. Finally, Chen et al. (2010) find that market participants reward

firms meeting or beating analysts' forecasts at the earnings announcement date no matter if accruals or real activities are managed to achieve analysts' forecasts.

We contribute to this stream of literature analyzing whether firms with signals of having engaged in earnings management to meet or beat the benchmarks are penalized by market participants. We predict that price-earnings multiples are lower for firms achieving a pattern of increasing earnings either through accrual-based earnings management and/or through real activities management. This leads to our second set of hypotheses:

**H2** (a): Investors reduce price-earnings multiples to firms achieving a pattern of increasing earnings through the opportunistic use of accruals.

**H2** (b): Investors reduce price-earnings multiples to firms achieving a pattern of increasing earnings through real activities management, including accelerating sales on credit beyond whatever is advisable by common practice, overproduction and opportunistically reducing discretionary expenses.

#### 3. Research Design

In our empirical tests, we choose a five-year pattern of increasing earnings as a cutoff based on the results in Barth et al. (1999) and the basic results do not change when we use different lengths of patterns in earnings increases.

#### 3.1 Estimation model

We conduct our analysis by estimating the following regression:

$$\begin{split} PRICE_{i,t} = \quad \beta_0 + \; \beta_1 \; EPS_{i,t} + \beta_2 \; EPS_{i,t} * DBEAT5_{i,t} \\ + \; \beta_3 \; EPS_{i,t} * DBEAT5_{i,t} * DCFO5_{i,t} \\ + \; \beta_4 \; EPS_{i,t} * DBEAT5_{i,t} * D\_AM\_ACCR5_{i,t} \\ + \; \beta_5 \; EPS_{i,t} * DBEAT5_{i,t} * D\_RM\_SALE5_{i,t} \\ + \; \beta_6 \; EPS_{i,t} * DBEAT5_{i,t} * D\_RM\_PROD5_{i,t} \\ + \; \beta_7 \; EPS_{i,t} * DBEAT5_{i,t} * D\_RM\_DIXP5_{i,t} \end{split}$$

$$\begin{split} &+\beta_8 \, EPS_{i,t} * Growth5_{,i,t} + \beta_9 \, EPS_{i,t} * Leverage_{i,t} \\ &+\beta_{10} \, EPS_{i,t} * Evar_{i,t} + \beta_{11} \, BVS_{i,t} + \varepsilon_{i,t} \end{split} \tag{1}$$

Where i indexes the firm and t indexes the year. PRICE is share price at the fiscal year end (COMPUSTAT #199). EPS is income before extraordinary items, NIBE (COMPUSTAT #18), divided by number of common shares outstanding for basic EPS, (COMPUSTAT #54).

Throughout our tests and to identify firms that report a pattern of increasing earnings, we focus on income before extraordinary items, NIBE. Equation (1) includes an indicator variable DBEAT5<sub>it</sub> that takes value 1 if firm i continuously reports increasing positive earnings for five years, and 0 otherwise. We expect that  $\beta_2$  is positive, indicating that firms with a five-year pattern of increasing positive earnings have higher price-earnings multiples than other firms (Barth et al., 1999).

We refer to firms with a five-year pattern both in earnings increases and cash flows increases as non-suspect beaters. We create an indicator variable DCFO5<sub>it</sub> that captures whether a firm reports a pattern of increasing cash flows, regardless of whether the pattern of increasing cash flows is linked to a pattern of increasing in earnings. We identify non-suspect beaters using an interaction term DBEAT5<sub>it</sub> \* DCFO5<sub>it</sub> taking the value of 1 if firm i reports five-year patterns of increases in positive increasing earnings and in positive increasing cash flows, and 0 otherwise. Regarding whether there are additional rewards to non-suspect beaters, we expect coefficient  $\beta_3$  to be positive, implying that non-suspect beaters (DBEAT5<sub>it</sub> \* DCFO5<sub>it</sub> = 1) enjoy higher price-earnings multiples than the rest of firms with a pattern of increasing earnings (DBEAT5<sub>it</sub> \* DCFO5<sub>it</sub> = 0).

For the test of market rewards to firms suspect of having engaged in earnings management to achieve the earnings pattern, we expect that market participants assign

lower price-earnings multiples to suspect firms. The indicator variable D\_AM\_ACCR5<sub>i,t</sub> captures whether firms would fail to achieve the pattern of increasing earnings if discretionary accruals are not included in earnings. The interaction term DBEAT5<sub>i,t</sub> \* D\_AM\_ACCR5<sub>i,t</sub> captures whether earnings beaters (firms with a five-year pattern of consecutive earnings increases) would fail to achieve the pattern of increasing earnings if discretionary accruals were not included in earnings. We expect that suspect firms  $(DBEAT5_{i,t} * D_AM_ACCR5_{i,t} = 1)$  are penalized by market participants, leading to lower price-earnings multiples. If that is the case, we expect coefficient  $\beta_4$  to be negative. D\_RM\_SALE5<sub>i,t</sub> is an indicator variable indicating whether firms would fail to achieve the pattern of increasing earnings if the effect of sales manipulation were not included in earnings. The interaction term DBEAT5<sub>i,t</sub> \* D\_RM\_SALE5<sub>i,t</sub> captures whether earnings beaters would fail to achieve the pattern of increasing earnings if the effect of sales manipulation were not included in earnings. We expect coefficient  $\beta_5$ to be negative, implying those suspect firms (DBEAT5<sub>i,t</sub> \* D\_RM\_SALE5<sub>i,t</sub> = 1) are penalized by market participants. We use an indicator variable D\_RM\_PROD5<sub>i,t</sub> to capture whether firms would fail to sustain a five-year pattern of increasing earnings if the effect of overproduction were not included in earnings. The interaction term DBEAT5<sub>i,t</sub> \* D\_RM\_PROD5<sub>i,t</sub> captures whether earnings beaters would fail to achieve the earnings pattern when the influence of overproduction were not included in earnings. We expect that coefficient  $\beta_6$  is negative, implying that market participants reduce price-earnings multiples to those suspect firm. Finally, the indicator variable D\_RM\_DIXP5<sub>i,t</sub> captures whether firms would fail to report a five-year pattern of increasing earnings if the effect of the opportunistic reduction of discretionary expenses were not included in earnings. The interaction term DBEAT5<sub>i,t</sub> \* D\_RM\_DIXP5<sub>i,t</sub> captures whether earnings beaters would fail to achieve the earnings pattern if the

opportunistic reduction of discretionary expenses were not included in earnings. We expect coefficient  $\beta_7$  to be negative, indicating that market participants reduce price-earnings multiples to firms suspect of having engaged in discretionary expenses management.

Finally, following Barth et al. (1999) we add four controls: Growth5, Leverage, Evar and BVS. Growth5 is the five-year compound growth rate of book value of equity (COMPUSTAT #60). We predict  $\beta_8$  to be positive. Leverage is defined as the sum of short-term debt due within one year and long-term debt, divided by market value of equity (COMPUSTAT #34 + COMPUSTAT #9) /(COMPUSTAT #199 \* COMPUSTAT #25). Evar is measured as the variance of the past five years' percentage change in earnings (NIBE<sub>i,t</sub> – NIBE<sub>i,t-1</sub>)/abs(NIBE<sub>i,t-1</sub>). Leverage is a measure of financial risk and Evar is a measure of operating risk. We expect  $\beta_9$  and  $\beta_{10}$  to be negative. BVS is book value of equity per share. Following Ohlson (1995) and Barth et al. (1999), we expect coefficient  $\beta_{11}$  to be positive.

# 3.2 Identifying suspect firms

We refer to firms with signals of having engaged in accrual-based earnings management or real activities management to sustain five years of consecutive earnings increases as suspect firms. We consider suspect firms those with (a) five years of consecutive earnings increases that present discretionary accruals that, if not included in earnings, would fail to report the pattern of increasing earnings, or (b) five years of consecutive earnings increases that engage in inefficient operational practices with the objective of managing earnings, and that would have broken the pattern of increasing earnings in the absence of those inefficient operational practices.

# 3.2.1 Identification of firms suspect of accrual-based earnings management

We use an indicator variable D\_AM\_ACCR5<sub>i,t</sub> taking the value of 1 if firm i's earnings without discretionary accruals are less than previous year's actual earnings during any of the five years, and 0 otherwise. We define an interaction term DBEAT5<sub>i,t</sub> \* D\_AM\_ACCR5<sub>i,t</sub> that takes value 1 if earnings beaters (firms with a five-year pattern of increasing earnings) would fail to report the earnings pattern in the absence of discretionary accruals in the reported earnings, and 0 otherwise. We obtain earnings without discretionary accruals by subtracting the abnormal accruals from the reported earnings, in which abnormal accruals are the difference between actual accruals and fitted normal accruals estimated using the Jones (1991) model.

The Jones (1991) model we use to estimate the normal level of accruals is as follows:

$$\frac{TA_{i,t}}{A_{i,t-1}} = \alpha_0 + \alpha_1 \left(\frac{1}{A_{i,t-1}}\right) + \alpha_2 \left(\frac{\Delta SALE_{i,t}}{A_{i,t-1}}\right) + \alpha_3 \left(\frac{PPE_{i,t}}{A_{i,t-1}}\right) + \varepsilon_{i,t}$$
 (2)

where TA<sub>i,t</sub> (Total accruals) is measured as income before extraordinary items, NIBE, minus cash flows from operations, CFO, (COMPUSTAT #18 – #308). The variable ΔSALE<sub>i,t</sub> is the change in sales revenues and PPE<sub>i,t</sub> is firm's gross property, plant, and equipment (COMPUSTAT #7). All variables are scaled by lagged total assets, A<sub>i,t-1</sub> (COMPUSTAT #6). A higher value of abnormal total accruals implies that managers are more likely to engage in income-increasing earnings management.

# 3.2.2 Identification of firms suspect of real earnings management

Firms suspect of having engaged in real activities management to achieve a five-year pattern of increasing earnings are those firms that undertake inefficient operating practices with the objective of avoiding breaking the pattern of increasing earnings. Following prior studies by Dechow et al. (1998) and Roychowdhury (2006), we consider three strategies of real activities management: accelerating sales on credit beyond whatever is advisable by common practice, producing more goods than

necessary and the opportunistic reduction of discretionary expenses.

# Sales manipulation

We create an indicator variable D\_RM\_SALE5<sub>i,t</sub> that takes value of 1 if firm i's earnings without the influence of sales management are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. We use an interaction indicator DBEAT5<sub>i,t</sub> \* D\_RM\_SALE5<sub>i,t</sub> taking value 1 if earnings beaters fail to maintain a five-year pattern of increasing earnings when the inefficient sales manipulation is not included in the calculation of earnings, and 0 otherwise. We generate earnings without the effect of sales manipulation by adding abnormal CFO to the reported earnings. The abnormal CFO is actual CFO minus the CFO that one would expect given sales.

Following Dechow et al. (1998), Roychowdhury (2006), Cohen et al. (2008), Bartov and Cohen (2009) and Cohen and Zarowin (2010), we express normal CFO from sales as a linear function of sales and the change in sales. Firms can accelerate sales to increase current earnings by offering price discounts or more lenient credit terms. The increased sales as a result of the price discounts and lenient credit terms are likely to disappear once the prices revert to the old ones. Such sales manipulation leads to higher current earnings when the sales are booked and margins are positive, however, it leads to lower current CFO given the normal sales levels (Roychowdhury, 2006). To estimate normal CFO, we run the following cross-sectional regression for each industry-year:

$$\frac{\mathit{CFO}_{i,t}}{\mathit{A}_{i,t-1}} = \alpha_0 + \alpha_1 \left(\frac{1}{\mathit{A}_{i,t-1}}\right) + \alpha_2 \left(\frac{\mathit{SALE}_{i,t}}{\mathit{A}_{i,t-1}}\right) + \alpha_3 \left(\frac{\Delta \mathit{SALE}_{i,t}}{\mathit{A}_{i,t-1}}\right) + \varepsilon_{i,t} \tag{3}$$

where  $CFO_{i,t}$  is cash flows from operation (COMPUSTAT #308). A more negative value of abnormal CFO implies that managers are more likely to engage in sales manipulation to increase earnings.

# Overproduction

We construct an indicator variable D\_RM\_PROD5<sub>i,t</sub> that equals 1 if firm i's earnings without the effect of overproduction are less than previous year's actual earnings during any of the five years, and 0 otherwise. We create an interaction term DBEAT5<sub>i,t</sub> \* D\_RM\_PROD5<sub>i,t</sub> taking value of 1 if earnings beaters break a five-year pattern of increasing earnings when the effect of overproduction is not included in earnings, and 0 otherwise.

Overproduction takes place when managers produce more goods than needed to report lower cost of goods sold (COGS) and therefore to increase current earnings. To capture and quantify overproduction, we estimate normal production costs using the following regression (e.g., Roychowdhury, 2006; Cohen et al., 2008; Cohen and Zarowin, 2010; Chen et al., 2010; Gunny 2010; Zang, 2012):

$$\frac{PROD_{i,t}}{A_{i,t-1}} = \alpha_0 + \alpha_1 \left(\frac{1}{A_{i,t-1}}\right) + \alpha_2 \left(\frac{SALE_{i,t}}{A_{i,t-1}}\right) + \alpha_3 \left(\frac{\Delta SALE_{i,t}}{A_{i,t-1}}\right) + \alpha_4 \left(\frac{\Delta SALE_{i,t-1}}{A_{i,t-1}}\right) + \varepsilon_{i,t} \tag{4}$$

where PROD<sub>i,t</sub> is production costs defined as the sum of COGS<sub>i,t</sub> (COMPUSTAT #44) and change in inventory (COMPUSTAT #3) during the year. The abnormal production costs are the difference between actual production costs and normal production costs. A higher value of abnormal production costs implies that managers are more likely to overproduce to increase earnings. We subtract abnormal production costs from earnings to obtain earnings without the effect of overproduction.

Discretionary expenses management (opportunistic decreases in discretionary expenses)

An indicator variable D\_RM\_DIXP5<sub>i,t</sub> equals 1 if firm i's earnings without the influence of opportunistic decreases in discretionary expenses are less than previous year's actual earnings during any of the five years, and 0 otherwise. We create an

interaction term DBEAT5<sub>i,t</sub> \* D\_RM\_DIXP5<sub>i,t</sub> that takes value 1 if earnings beaters break a five-year pattern of increasing earnings when the effect of discretionary expenses management is not included in earnings, and 0 otherwise.

Discretionary expenses are generally expensed in the period when they are incurred; however, firms can opportunistically reduce current discretionary expenses to inflate current earnings. We estimate normal levels of discretionary expenses using the following model (e.g. Roychowdhury, 2006; Cohen et al., 2008; Cohen and Zarowin, 2010):

$$\frac{DIXP_{i,t}}{A_{i,t-1}} = \alpha_0 + \alpha_1 \left(\frac{1}{A_{i,t-1}}\right) + \alpha_2 \left(\frac{SALE_{i,t-1}}{A_{i,t-1}}\right) + \varepsilon_{i,t}$$

$$(5)$$

where DIXP<sub>i,t</sub> is discretionary expenses, including R&D (COMPUSTAT #46), selling, general and administrative expenses (COMPUSTAT #189) and advertising expenses (COMPUSTAT #45). The abnormal discretionary expenses are the difference between actual discretionary expenses and normal discretionary expenses estimated with equation (5). We calculate earnings without the effect of discretionary expenses management by adding abnormal discretionary expenses to the reported earnings.

# 4. Sample, Descriptive Statistics and Results

#### 4.1 Sample

Our initial sample consists of all available firms on the annual industrial and research COMPUSTAT North America databases between 1990 and 2007, excluding regulated firms (SIC codes between 4400 and 4999), and banks and financial institutions (SIC codes between 6000 and 6999). We require at least fifteen observations for each industry-year to perform cross-sectional regressions to estimate normal levels of

accruals and operational activities (Roychowdhury, 2006; Zang, 2012). We drop observations with missing data of income before extraordinary items and/or number of common shares outstanding are missing. A few observations of EPS take on extreme values, so we eliminate the upper and lower 1% of the EPS for each year (Burgstahler and Dichev, 1997). This finally yields a sample of 83,443 firm-year observations from 1990 to 2007. This is the starting data set we use to estimate normal levels of total accruals in equation (2), and normal levels of the three types of real activities in equations (3), (4) and (5). We employ, though, restricted subsets in our main tests.

# 4.2 Descriptive statistics

Table 1 presents the number of firms with different patterns of increasing positive earnings and cash flows from 1990 to 2007. Columns 2 and 3 in Table 1 show the number of firms reporting different patterns of increasing positive earnings and different patterns of increasing positive cash flows. The number of firms with earnings patterns and cash flow patterns decreases with the length of patterns. More firms report a one-year pattern of increasing cash flows than a one-year pattern of increasing earnings. However, there are more firms reporting consecutive earnings increases than cash flow increases from a two-year pattern to a twelve-year pattern. Besides, the number of firms reporting cash flow patterns decreases more rapidly with the length of patterns than the number of firms reporting earnings patterns. This is preliminary evidence that firms mainly use earnings, but not cash flows, to signal to the market that their business follows a stable growth trend. This is in line with a survey study conducted by Graham et al. (2005, p.49), showing that Chief Financial Officers think earnings volatility matters more than cash flow volatility because "the market becomes

<sup>&</sup>lt;sup>1</sup> We estimate normal levels of total accruals and operating activities cross-sectionally with at least fifteen observations for each two-digit SIC industry-year between 1990 and 2007.

more skeptical of underlying cash flows when earnings are volatile." Executives believe that a firm is perceived as riskier by the market when it has more volatile earnings than another firm even if these two firms have the same cash flow volatility (Graham et al., 2005). This evidence is also in line with managers using additional tools such as accruals and/or operational activities to sustain their consecutive increases in earnings. Column 4 presents the number of firms that have the same patterns of increasing positive earnings and increasing positive cash flows. The number decays at a rate higher than 50% with the length of patterns.

To carry out our tests on whether market participants reward differently firms in line with their means of achieving a five-year pattern of increasing earnings, we first constrain the sample to firms with at least five years of earnings history and 39,275 profit- and loss-making firm-year observations remain. We require profit-making firms for our estimation sample, and this criterion eliminates 14,129 observations of negative earnings, yielding 25,146 observations. We further require an identical sample throughout all our tests to avoid empirical results varying across estimation models, and this requirement eliminates 2,541 observations, which yields our final sample of 22,605 profit-making firm-year observations between 1995 and 2007. We use this sample for testing all of our hypotheses. Table 1 also shows that out of the 2,088 earnings beaters (firms with a five-year pattern of increasing earnings), 18% (368) are non-suspect beaters (firms with five-year patterns of increasing earnings increases and increasing cash flows).

# [Insert Table 1]

Table 2 presents descriptive statistics based on the sample for which we report regression results. Table 2 also presents means for earnings beaters (firms with a five-year pattern of increasing earnings) and non-earnings beaters (firms without a five-year pattern of increasing earnings) separately, and t-tests for differences in means across the two subsamples. Specifically, earnings beaters have significantly higher share price, earnings, EPS and CFO than non-earnings beaters. Non-earnings beaters have significantly lower sales, total assets, market capitalization and ratio of market value of equity to book value of equity than earnings beaters.

Total accruals scaled by lagged total assets (TA/A) are similar for earnings beaters and for non-earnings beaters and not significantly different. CFO scaled by lagged total assets (CFO/A) is 0.16 for earnings beaters versus 0.12 for non-earnings beaters, and the difference is significant. The value of production costs scaled by lagged total assets (PROD/A) is 1.15 for the earnings beaters and it is significantly higher than scaled production costs for non-earnings beaters (0.99). The scaled discretionary expenses (DIXP/A) are not significantly different across the two subsamples. In addition, earnings beaters have significant higher growth in book value of equity (Growth5) and lower financial risk (Leverage) than non-earnings beaters.

# [Insert Table 2]

Table 3 presents Pearson correlations between various variables. Share price (PRICE) exhibits strong positive correlation with earnings, EPS, CFO, scaled CFO, abnormal CFO (Ab\_CFO/A), Growth5 and BVS, and significantly negative correlation with scaled production costs, scaled discretionary expenses, abnormal production costs

(Ab\_PROD/A) and Leverage. Consistent with prior studies, earnings are positively correlated with CFO (0.93) and scaled CFO (0.03), and negatively correlated with scaled production costs, scaled discretionary expenses and Leverage. EPS is negatively correlated with abnormal production costs. As expected, the scaled CFO and scaled total accruals exhibit a strong negative correlation, with a correlation coefficient of -0.96.

The correlation between abnormal accruals (Ab\_TA/A) and abnormal CFO is significantly negative (-0.11). This is probably because accrual-based earnings management and sales manipulation take place simultaneously (Roychowdhury, 2006; Zang, 2012). The correlation coefficient between abnormal production costs and abnormal discretionary expenses (Ab\_DIXP/A) is significantly negative (-0.33). This is, probably, because managers engage in overproduction, leading to abnormally high production costs; meanwhile, they reduce discretionary expenses when the common goal is to report higher earnings (Roychowdhury, 2006).

#### [Insert Table 3]

# 4.3 Empirical results

We estimate equation (1) using standard errors clustered by firm and year to control for serial correlation and cross-sectional dependence (Petersen, 2009) for 22,605 profit-making firm-year observations from 1995 to 2007.<sup>2</sup> We include year and industry

2

<sup>&</sup>lt;sup>2</sup> We also estimate equation (1) using White (1980) standard errors. Results are very similar. The t-statistics are greater if we use White (1980) standard errors rather than Petersen (2009) standard errors, but inferences do not change.

dummies to control for fixed effects.

Table 4 reports the regression-based tests of hypothesis H1 regarding whether investors assign higher price-earnings multiples to non-suspect beaters (firms with fiveyear patterns both in earnings increases and cash flow increases). The first Column shows baseline results of basic price-earnings multiples and other factors that theory suggests and prior empirical work has shown to have significant effects on share price. The results are consistent with our predictions regarding significantly positive coefficients on growth of book value of equity (EPS \* Growth5) (4.993, t = 2.88) and on book value of equity per share (BVS) (1.258, t = 3.60) and significantly negative coefficients on earnings variability (EPS \* Evar) (0, t = -1.80) and on leverage ratio (EPS \* Leverage) (-2.481, -5.98). Column 2 shows that the coefficient on the test variable EPS \* DBEAT5 is significantly positive (2.940, t = 4.51), consistent with Barth et al. (1999) that firms with a five-year pattern of increasing earnings have higher price-earnings multiples than other firms.<sup>3</sup> In Column 3 the price-earnings multiple of DBEAT5 is 2.398 (t = 3.86) and the price-earnings multiple of the interaction term DBEAT5 \* DCFO5 is 3.696 (t = 2.74), indicating that market participants add additional rewards to non-suspect beaters, consistent with our hypothesis H1.

# [Insert Table 4]

Table 5 shows regression results of whether investors assign lower price-earnings multiples to firms that engage in several earnings management types to achieve earnings

<sup>3</sup> Although a five-year pattern is an arbitrary choice, untabulated tests show that market participants also assign higher price-earnings multiples to firms that have consecutive patterns of earnings increases from two years through eleven years.

patterns. Column 1 in Panel A shows that the coefficient on EPS \* DBEAT5 is significantly positive (3.483, t = 4.70), meaning that market participants assign higher price-earnings multiple to earnings beaters. The coefficient on EPS \* DBEAT5 \* D AM ACCR5 is significantly negative (-2.203, t = -2.85), indicating that market participants assign lower price-earnings ratios to suspect firms that fail to maintain five years of consecutive increases in earnings if discretionary total accruals are not included in earnings, which is consistent with hypothesis H2 (a).<sup>4</sup> Column 2 shows that the coefficients on earnings beaters is 3.551 (t = 4.59) and that the interaction term EPS \* DBEAT5 \* D RM SALE5 is significantly negative (-3.821, t = -3.46), meaning that market participants penalize suspect firms that would have broken the pattern of increasing earnings if the effect of sales manipulation on earnings was not considered, consistent with hypothesis H2 (b). Column 3 shows that market participants reward earnings beaters (3.418, t = 4.40) and the coefficient on the interaction term EPS \* DBEAT5 \* D\_RM\_PROD5 is significantly negative (-1.584, t = 2.12), consistent with our hypothesis H2 (b) that firms that overproduce to avoid breaking the earnings trend have lower price-earnings multiples than the rest of benchmark beating firms. In Column 4 the coefficient on EPS \* DBEAT5 \* D RM DIXP5 is insignificant (-0.325, t = -0.33), implying that market participants do not penalize firms that opportunistically reduce discretionary expenses to achieve the earnings pattern, which is inconsistent with our hypothesis H2 (b).

Given that accrual-based earnings management and real activities earnings management probably take place at the same time (Barton, 2001; Zang, 2012), we estimate equation (1) including all earnings management proxies at the same time.

4

<sup>&</sup>lt;sup>4</sup> This result also holds if we use discretionary working capital accruals.

Table 5 Panel B presents regression results of whether market participants assign lower price-earnings multiples to suspect firms that have used both accrual-based earnings management and each type of real activities management to increase earnings simultaneously. Column 1 in Panel B shows that earnings beaters are rewarded (4.092, t = 4.62) and that market participants reduce price-earnings multiples to suspect firms that have managed total accruals (-2.197, t = -2.69) and sales (-3.816, t = -3.47) to increase earnings. Column 2 shows that market participants reward earnings beaters (3.864, t = 4.56) and reduce price-earnings multiples to suspect firms that use accrual-based earnings management (-2.050, t = -2.68) together with overproduction (-1.385, t = -1.82). Column 3 shows that price-earnings multiples are lower for suspect firms that employ accrual-based earnings management (-2.214, t = -3.08) but not for firms with opportunistic decreases in discretionary expenses (0.048, t = 0.05). Finally, Column 4 shows that market participants do not reduce rewards to firms suspect of overproduction (-1.008, t = -1.48) when we consider all types of earnings management. This result could be explained by the overlapping classification of different types of suspect firms.

# [Insert Table 5]

We carry out an additional test to check whether the results in Table 5 Panel A do not vary when the proxy for non-suspect beaters is also included in the models. Table 6 contains regression results of market rewards on earnings beaters, non-suspect beaters and each type of suspect firms. Column 1 shows that the coefficient on EPS \* DBEAT5 \* DCFO5 is 3.567 (t = 2.61) and the coefficient on EPS \* DBEAT5 \* D\_AM\_ACCR5 is -2.050 (t = -2.66), implying that market participants assign higher

price-earnings multiples to non-suspect beaters and lower price-earnings multiples to suspect firms that manage accruals. Column 2 reports that the coefficient on EPS \* DBEAT5 \* DCF05 is 3.114 (t = 2.44) and the coefficient on EPS \* DBEAT5 \* D RM SALE5 is -3.298 (t = -3.16), indicating that market participants assign additional price-earnings multiples to non-suspect beaters and reduces price-earnings multiples to suspect firms that use sales manipulation to report a pattern of increasing earnings. Column 3 shows that market participants assign higher rewards to non-suspect beaters (3.570, t = 2.66) and penalize suspect firms that overproduce to avoid breaking the earnings pattern (-1.389, t = -1.83). Column 4 shows that the coefficient on EPS \* DBEAT5 \* DCFO5 is 3.728 (t = 2.73) and the coefficient on EPS \* DBEAT5 \* D RM DIXP5 is insignificant (-1.389, t = -1.83), indicating that market participants do not penalize suspect firms that reduce discretionary expenses opportunistically to avoid breaking the earnings trend. The regression results in Table 6 are consistent with the preceding findings in Tables 4 and 5 that market participants assign higher priceearnings multiples to non-suspect beaters and lower price-earnings multiples to suspect firms that would have broken a five-year pattern of increasing earnings if discretionary accruals, the effects of sales manipulation or the effects of overproduction were not included in the reported earnings.<sup>5</sup>

# [Insert Table 6]

#### 5. Robustness Tests

<sup>&</sup>lt;sup>5</sup> We repeat the analysis excluding BVS. Our main inferences do not change.

# 5.1 Earnings increases not sustained by sales growth

Ghosh et al. (2005) show that firms reporting consecutive earnings increases together with sales growth have higher price-earnings multiples than firms reporting earnings increases alone. Given that sales are more difficult to manipulate than expenses (Ertimur et al., 2003; Ghosh et al., 2005), and continuous earnings increases supported by sales growth are perceived as higher earnings quality by the market (Ghosh et al., 2005), we use sales growth as a signal to gauge whether the pattern of increasing earnings has been fabricated. In particular, we consider firms that report earnings increases but that do not report increases in sales as suspect firms. We expect that market participants assign lower price-earnings multiples to these suspect firms.

To carry out our tests, we create an indicator variable  $D_NONREV5_{i,t}$  taking the value of 1 if firm i does not report increasing sales revenues for five years, and 0 otherwise. We define an interaction term  $DBEAT5_{i,t} * D_NONREV5_{i,t}$  that takes value 1 if firm i reports a five-year pattern of earnings increases but not a five-year pattern of increasing sales, and 0 otherwise.

Table 7 Panel A shows that of the 2,088 earnings beaters (firms with a five-year pattern of earnings increases), 19% (398) do not report a five-year pattern of sales growth. We view this group of earnings beaters as suspect firms that might have used earnings management to maintain the earnings pattern. Panel B shows that the coefficient on  $EPS_{i,t} * DBEATS_{i,t}$  is significantly positive (3.530, t = 4.73), and the coefficient on  $EPS_{i,t} * DBEATS_{i,t} * D_NONREVS_{i,t}$  is significantly negative (-2.70, t = -3.80), meaning that market participants assign lower price-earnings multiples to suspect firms when their pattern of increasing earnings is not achieved through sales growth.

# [Insert Table 7]

# 5.2 Additional analyses

To verify the robustness of our main findings, we repeat the tests of hypotheses H1, H2 (a) and H2 (b) using (1) a sample of 39,275 profit- and loss-making firm-year observations, and (2) a sample of 2,088 firms with a pattern of five-year earnings increases.

The results with the sample of 39,275 profit- and loss- making firm-year observations (EPS≥0 and EPS<0) are not qualitatively different from our main results. Following Francis et al. (2003), we control for the effect of loss-making firm-year observations using an indicator variable, DLOSS<sub>i,t</sub>, taking the value of 1 if firm i's earnings in year t is less than 0, and 0 otherwise. Unreported results show that earnings beaters are rewarded throughout all model specifications and non-suspect beaters are valued higher than the rest of earnings pattern beating firms. The results also show that market participants assign lower price-earnings multiples to firms suspect of having used accrual-based earnings management, sales manipulation or overproduction to achieve a five-year pattern of increasing earnings.

Second, we repeat the test for hypotheses H1, H2(a) and H2(b) using a sample of firms meeting or beating prior year earnings for five years (DBEAT5 = 1), which left 2,088 firm-year observations. The results are consistent with the evidence previously reported in our main tests that non-suspect beaters have higher price-earnings multiples and that suspect firms that manage discretionary accruals, sales and production have lower price-earnings multiples than the rest of earnings pattern beating firms.

# 6. Summary and Conclusion

Prior research (Francis et al., 2003; Burgstahler and Eames, 2003) provides mixed results as to whether market participants can see through managerial attempts at fabricating streams of increases in earnings. In this paper we complement the existing literature by analyzing whether market participants reward firms with obvious signals of not having engaged in earnings management to meet or beat prior year's earnings. As a signal that the earnings stream is genuine we look at whether the firm also reports a pattern of increases in cash flows. We classify firms with a stream of increases in both earnings and cash flows as Non-suspect firms, and find that market rewards to benchmark beating firms are more pronounced for these Non-suspect firms.

We also analyze whether market participants penalize firms suspect of fabricating the earnings stream, either through accrual-based earnings management or through the manipulation of real activities. We find that market participants assign lower price-earnings multiples to firms suspect of using accrual-based earnings management to achieve a five-year pattern of increasing earnings. Our results also show that market participants penalize firms that increase credit sales beyond whatever is advisable by common practice or overproduce to reduce cost of goods sold, to achieve the pattern of increasing earning. However, market participants do not penalize firms that achieve the earnings pattern through opportunistic reductions in discretionary expenses, including R&D, advertising, and selling, general and administrative expenses.

While we expected firms that opportunistically reduce discretionary expenses to be penalized, our empirical results about the opportunistic reduction of discretionary expenses are consistent with those in Bhojraj et al. (2009) that firms that meet analysts'

forecasts through real earnings management are not immediately punished in the market. Whether this result can be attributed to the limited attention argument discussed by Daniel et al., 2002 and Hirshleifer and Teoh, 2003, to other market imperfections, or to other more rational explanations, is something that we do not tackle in the present study, but that should be the object of future research. An alternative explanation for this result is that the proxy that we use to capture opportunistic decreases in discretionary expenses could be noisy and not used in practice by financial analysts.

#### References

- Athanasakou, V., N. C. Strong, and M. Walker, 2011. The market reward for achieving analyst earnings expectations: does managing expectations or earnings management matter? *Journal of Business Finance & Accounting* 38: 58-94.
- Barth, M. E., J. A. Elliott and M. Finn, 1999. Market rewards associated with increasing earnings patterns. *Journal of Accounting Research* 37 (2): 387-414.
- Barton, J. 2001. Does the use of financial derivatives affect earnings management decision? *The Accounting Review* 76: 1-26.
- Bartov, E., and D. A. Cohen, 2009. The "numbers game" in the pre- and post-sarbanes-oxley eras. *Journal of Accounting, Auditing and Finance* 24: 505-534.
- \_\_\_\_\_, D. Givoly, and C. Hayn, 2002. The rewards to meeting or beating earnings expectations'. *Journal of Accounting and Economics* 33: 173-204.
- Bhojraj, S., P. Hribar, M. Picconi and J. McInnis, 2009. Making sense of cents: An examination of firms that marginally miss or beat analyst forecasts. *The Journal of Finance* 64(5): 2361-2388.
- Burgstahler, D., and I. Dichev, 1997. Earnings management to avoid earnings decreases and losses. *Journal of Accounting and Economics* 24 (1): 99-126.
- \_\_\_\_\_, and M. J. Eames, 2003. Earnings management to avoid losses and earnings decreases: are analysts fooled? *Contemporary Accounting Research* 20 (2): 253-294.
- Chen, J. Z., L. Rees, and K. Sivaramakrishnan, 2010. On the use of accounting vs. real earnings management to meet earnings expectations a market analysis. (available at SSRN: http://ssrn.com/abstract=1070122).
- Cohen, D.A., A. Dey, and T.Z. Lys, 2008. Real and accrual-based earnings management in the pre- and post-Sarbanes-Oxley periods. *The Accounting Review* 83 (3): 757-787.
- \_\_\_\_\_, and P. Zarowin, 2010. Accrual-based and real earnings management activities around seasoned equity offerings. *Journal of Accounting and Economics* 50: 2-19.
- Daniel, K.D., D. Hirshleifer, and S. H. Teoh, 2002. Investor psychology in capital markets: evidence and policy implications. *Journal of Monetary Economics* 49 (1): 139-209.
- Dechow, P., S. P. Kothari, and R. L. Watts, 1998. The relation between earnings and cash flows. *Journal of Accounting and Economics* 25: 133-168.
- Degeorge, F., J. Patel, and R. Zeckhauser, 1999. Earnings management to exceed thresholds. *Journal of Business* 72 (1): 1-33.
- Ertimur, Y., J. Livnat, and M. Martikainen, 2003. Differential market reaction to revenue and expense surprises. *Review of Accounting studies* 8: 185-211.
- Francis, J., R. LaFond, P. Olsson, and K. Schipper, 2003. Earnings quality and the pricing effects of earnings patterns. *Working Paper*. The Fuqua School of Business, Duke University.
- Ghosh, A., Z., Gu, and P. C. Jain, 2005. Sustained earnings and revenue growth, earnings quality, and earnings response coefficients. *Review of Accounting Studies* 10: 33-57.

- Graham, J., C. Harvey, and S. Rajgopal, 2005. The economic implications of corporate financial reporting. *Journal of Accounting and Economics* 40: 3-73.
- Gunny, K. A., 2010. The relation between earnings management using real activities manipulation and future performance: evidence from meeting earnings benchmarks. *Contemporary Accounting Research* 27 (3): 855-888.
- Hirshleifer, D., and S. H. Teoh, 2003. Limited attention, information disclosure, and financial reporting. *Journal of Accounting and Economics* 36: 337-386.
- Jones, J., 1991. Earning management during import relief investigations. *Journal of Accounting Research* 29 (2): 193-228.
- Kasznik, R., 1999. On the association between voluntary disclosure and earnings management. *Journal of Accounting Research* 37 (1): 57-81.
- \_\_\_\_\_, and M. McNichols, 2002. Does meeting earnings expectations matter? evidence from analyst forecast revisions and share prices. *Journal of Accounting Research* 40 (3): 727-759.
- Koonce, L. and M. G. Lipe, 2010. Earnings trend and performance relative to benchmarks: how consistency influences their joint use. *Journal of Accounting Research* 48 (4): 859-884.
- Myers, J. N., L. A. Myers, and S. J. Douglas, 2007. Earnings momentum and earnings management. *Journal of Accounting, Auditing & Finance* 22(2): 249-284.
- Ohlson, J. A., 1995. Earnings, book values, and dividends in equity valuation. *Contemporary Accounting Research* 11(2): 661-687.
- Petersen, M. A., 2009. Estimating standard errors in finance panel data sets: comparing approaches. *The Review of Financial Studies* 22: 435-480.
- Roychowdhury, S., 2006. Earnings management through real activities manipulation. *Journal of Accounting and Economics* 42: 335-370.
- White, H., 1980. A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity. *Econometrica* 48(4): 817-838.
- Zang, A., 2012. Evidence on the tradeoff between real manipulation and accrual-based earnings management. *The Accounting Review* 87 (2): 675-703.

Table 1 Number of Firms Reporting Patterns of Positive Earnings Increases, Positive CFO increases and Both in Positive Earnings and Positive CFO increases

Num. of	Num. of	Num. of	Num. of Non-suspect Beaters				
Years	Earnings Beaters	CFO Beaters	(% over Earnings Beaters)				
1	21995	22468	12791(0.58)				
2	11369	9162	4428(0.39)				
3	6292	4070	1806(0.29)				
4	3601	1973	813(0.23)				
5	2088	993	368(0.18)				
6	1246	535	189(0.15)				
7	767	291	105(0.14)				
8	484	157	61(0.13)				
9	313	82	33(0.11)				
10	203	41	14(0.07)				
11	136	19	3(0.02)				
12	95	9	0(0)				

Earnings = net income before extraordinary items (Compustat #18). CFO = cash flows from operations (Compustat #308).

Table 2
Descriptive Statistics of Key Variables

Full S	Sample, 1995- (n = 22,605)	2007		Earnings Beaters (n = 2,088)	Non-earnings Beaters $(n = 20,517)$	t-statistic		
Variable	Mean	Median	Std. Dev.	Mean	Mean	(Diff. in Mean)		
PRICE	19.072	14.20	22.56	27.50	18.22	18.05***		
Earnings	214.74	19.23	981.01	480.58	187.69	13.05***		
EPS	1.04	0.74	1.10	1.46	0.99	18.69***		
CFO	368.42	30.94	1622.97	720.10	332.51	10.42***		
Total Accruals	-153.78	-9.11	793.16	-236.69	-145.31	-5.00***		
Sales	3259.81	390.58	12921.84	6582.32	2921.68	12.37***		
Total Assets	3440.72	349.77	17648.40	7291.31	3048.86	10.24***		
Market Capitalization	4517.63	393.84	19808.96	11384.65	3816.87	16.73***		
Market to Book	3.17	2.08	12.94	4.42	3.04	4.65***		
TA/A	-0.03	-0.04	0.27	-0.03	-0.03	0.44		
CFO/A	0.12	0.11	0.27	0.16	0.12	7.62***		
PROD/A	1.01	0.80	0.96	1.15	0.99	6.87***		
DIXP/A	0.35	0.28	0.32	0.35	0.35	1.04		
Ab_TA/A	0.08	0.05	0.21	0.06	0.08	-4.27***		
Ab_CFO/A	0.12	0.09	0.18	0.14	0.12	4.63***		
Ab_PROD/A	0.03	-0.06	0.62	-0.03	0.04	-4.94***		
Ab_DISEXP/A	-0.13	-0.10	0.30	-0.16	-0.13	-4.34***		
Leverage	0.39	0.15	1.07	0.17	0.41	-9.83***		
Evar	945.35	1.54	32101.12	10.21	1040.52	-1.397		
Ggrowth5	0.17	0.13	0.24	0.27	0.16	21.50***		
BVS	8.43	6.47	8.85	8.13	8.09	0.20		

Number of observations: 22,605 firm-year observations during 1995-2007. \*/\*\*/\*\*\* indicate Significance at the 10%/5% /1%.

PRICE = close price per share at the fiscal year end (Compustat #199). Earnings = net income before extraordinary items (Compustat #18). EPS = net income before extraordinary items divided by number of shares outstanding for basic EPS (Compustat #54). CFO = cash flows from operations (Compustat #308). Total Accruals = the difference between net income before extraordinary items and cash flows from operations. Sales = net sales (Compustat #12). Total Assets = Compustat #6). Market capitalization = close price per share times the number of common shares outstanding (Compustat #25). Market to Book = the market capitalization divided by the book value of common equity (Compustat

#### Table 2 (continued)

#60). TA/A = total accruals divided by lagged total assets. CFO/A = cash flows from operations divided by lagged total assets. PROD/A = Production costs divided by lagged total assets, where production costs are defined as the sum of cost of goods sold (Compustat #41) and the change in inventories (Compustat #3). DIXP/A = Discretionary expenses divided by lagged total assets, where discretionary expenses are the sum of R&D expenses (Compustat #46), advertising expenses (Compustat #45) and SGA expenses (Compustat #189). Ab\_TA/A = the discretionary total accruals computed using the Jones Model. Ab\_CFO/A = the level of abnormal cash flows from operations computed using equation (3). Ab\_PROD/A = the level of abnormal production costs computed using equation (4). Ab\_DIXP/A = the level of abnormal discretionary expenses computed using equation (5). Leverage = sum of short-term debt (Compustat #34) and long-term debt (Compustat #9) divided by market capitalization. Evar = variance of the past 5 years' percentage change in earnings. Growth5= 5-year compound growth rate of book value of equity. BVS = book value of equity per share.

Table 3
Correlation Matrix among Key Variables

	Firm characteristics							Abnormal earnings management					Controls			
-	PRICE	Earnings	EPS	CFO	TA/A	CFO/A	PROD/A	DIXP/A	Ab_T	A/A A	.b_CFO/A	Ab_PROD/A At	DIXP/A	Leverage	Evar	Growth5
Earnings	0.21**	1														
EPS	$0.62^{**}$	$0.27^{**}$	1													
CFO	$0.20^{**}$	$0.93^{**}$	$0.23^{**}$	1												
TA/A	-0.02	-0.01	0.01	-0.03**	1											
CFO/A	$0.04^{**}$	$0.03^{**}$	$0.05^{**}$	$0.03^{**}$	-0.96 **	1										
PROD/A	-0.08**	-0.06**	0.01	-0.07**	0.03 **	-0.02	1									
DIXP/A	-0.08**	-0.06**	-0.16**	-0.08**	0.02	0.04 **	0.13	1								
$Ab_TA/A$	-0.02	-0.01	-0.02	-0.02**	-0.27 **	-0.18 **	-0.03 **	0.05 **								
Ab_CFO/A	$0.06^{**}$	$0.07^{**}$	-0.02	$0.07^{**}$	-0.13 **	0.20 **	-0.27 **	0.07 **	-0	.11**	1					
Ab_PROD/A	-0.09**	-0.06**	-0.03**	-0.07**	0.02	-0.03 **	0.71 **	-0.01	0	.03**	-0.25**	1				
Ab_DIXP/A	-0.01	-0.01	-0.01	-0.02	-0.04 **	0.05 **	-0.16 **	0.48	0	.03**	-0.26**	-0.33**	1			
Leverage	-0.10**	-0.03**	-0.03**	0.01	0.01	-0.06 **	0.05 **	-0.11 **	-0	.01	-0.12**	0.02	-0.00	1		
Evar	-0.01	-0.01	-0.01	-0.01	0.01	-0.01	0.01	0.00	0	.00	-0.02	0.01	0.01	-0.00	1	
Growth5	$0.10^{**}$	0.01	$0.03^{**}$	-0.00	-0.00	0.04 **	-0.02	-0.03 **	-0	.01	$0.10^{**}$	-0.09**	-0.03**	-0.09**	0.11	1
BVS	$0.69^{**}$	$0.09^{**}$	$0.60^{**}$	$0.11^{**}$	-0.01	-0.03 **	-0.03 **	-0.20 **	-0	.04**	-0.09**	-0.06**	-0.02	0.01	-0.01	$0.05^{**}$

Number of observations: 22,605 firm-year observations during 1995-2007. \*\* represents the correlation coefficients are significant at the 5% level.

PRICE = close price per share at the fiscal year end (Compustat #199). Earnings = net income before extraordinary items divided by number of shares outstanding for basic EPS (Compustat #54). CFO = cash flows from operations (Compustat #308). TA/A = total accruals divided by lagged total assets. CFO/A = cash flows from operations divided by lagged total assets. PROD/A = Production costs divided by lagged total assets, where production costs are defined as the sum of cost of goods sold (Compustat #41) and the change in inventories (Compustat #45) and SGA expenses (Compustat #48). Ab\_TA/A = the discretionary expenses are the sum of R&D expenses (Compustat #46), advertising expenses (Compustat #41) and the change in inventories (Compustat #41) and the change in inventories (Compustat #3). Ab\_DIXP/A = the level of abnormal production costs computed using equation (4), where production costs are defined as the sum of cost of goods sold (Compustat #41) and the change in inventories (Compustat #3). Ab\_DIXP/A = the level of abnormal discretionary expenses, where discretionary expenses are the sum of R&D expenses (Compustat #46), advertising expenses (Compustat #45) and SGA expenses (Compustat #189). Leverage = sum of short-term debt (Compustat #34) and long-term debt (Compustat #34) divided by market capitalization. Evar = variance of the past 5 years' percentage change in earnings. Growth5 = 5-year compound growth rate of book value of equity. BVS = book value of equity per share.

Table 4
Analysis of Market Rewards to Non-suspect Beaters (Firms with Five-year Patterns of Increasing Earnings and Increasing Cash Flows)

N=22,605		1	2	3
Variable	Predicted	Coef.	Coef.	Coef.
variable	Sign	(t-statistic)	(t-statistic)	(t-statistic)
Intercept	?	1.061	1.057	1.047
		(0.99)	(0.97)	(0.96)
EPS	+	6.228***	5.749**	5.756**
		(3.73)	(3.37)	(3.37)
EPS*DBEAT5	+		2.940***	2.398***
			(4.51)	(3.86)
EPS*DBEAT5*DCFO5	+			3.696**
				(2.74)
EPS*Growth5	+	4.993***	4.792**	4.741**
		(2.88)	(2.46)	(2.40)
EPS*Leverage	-	-2.481***	-2.389***	-2.382***
C		(-5.98)	(-5.80)	(-5.80)
EPS*Evar	-	$-0.000^*$	-0.000	-0.000
		(-1.80)	(-1.59)	(-1.58)
BVS	+	1.258***	1.279***	1.280***
		(3.60)	(3.65)	(3.65)
R-Squared		0.574	0.579	0.580

PRICE = close price per share at the fiscal year end (Compustat #199). EPS = net income before extraordinary items divided by number of shares outstanding for basic EPS (Compustat #54). DBEAT5 = indicator variable equals 1 if a firm reports a 5-year pattern of increasing positive earnings, and 0 otherwise. DCFO5 = indicator variable equals 1 if a firm reports a 5-year pattern of increasing positive cash flows, and 0 otherwise. DBEAT5\*DCFO5 = interaction indicator variable equals 1 if a firm reports five-year patterns both in earnings increases and in cash flows increases, and 0 otherwise. Leverage = sum of short-term debt (Compustat # 34) and long-term debt (Compustat # 9) divided by market capitalization. Evar = variance of the past 5 years' percentage change in earnings. Growth5 = 5-year compound growth rate of book value of equity. BVS = book value of equity per share.

Table 5 Analysis of Market Rewards to Suspect Firms

Panel A: Having Used Each Type of Earnings Management to Achieve the Earnings Pattern					
N=22,605		1	2	3	4
Vaniahla	Predicted	Coef.	Coef.	Coef.	Coef.
Variable	Sign	(t-statistic)	(t-statistic)	(t-statistic)	(t-statistic)
Intercept	?	1.024	1.034	1.051	1.059
		(0.94)	(0.94)	(0.96)	(0.97)
EPS	+	5.751***	5.736***	5.757***	5.748***
		(3.37)	(3.36)	(3.37)	(3.36)
EPS*DBEAT5	+	3.483***	3.551***	3.418***	3.088***
		(4.70)	(4.59)	(4.40)	(3.27)
EPS*DBEAT5*D_AM_ACCR5	-	-2.203***			
		(-2.85)			
EPS*DBEAT5*D_RM_SALE5	-		-3.821***		
			(-3.46)		
EPS*DBEAT5*D RM PROD5	-			-1.584**	
				(-2.12)	
EPS*DBEAT5*D RM DIXP5	-				-0.325
					(-0.33)
EPS*Growth5	+	4.787**	4.801**	4.756**	4.785**
		(2.45)	(2.48)	(2.42)	(2.46)
EPS*Leverage	-	-2.377***	-2.369***	-2.392***	-2.389***
-		(-5.78)	(-5.84)	(-5.82)	(-5.80)
EPS*Evar	-	-0.000	-0.000	-0.000	-0.000
		(-1.58)	(-1.58)	(-1.59)	(-1.59)
BVS	+	1.279***	1.282***	1.279***	1.279***
		(3.65)	(3.66)	(3.65)	(3.65)
R-Squared		0.579	0.580	0.579	0.579

Table 5 (continued)

N=22,605 Variable	Predicted	1	2	3	4
Variable					4
	a.	Coef.	Coef.	Coef.	Coef.
	Sign	(t-statistic)	(t-statistic)	(t-statistic)	(t-statistic)
Intercept	?	1.001	1.021	1.023	1.003
		(0.91)	(0.93)	(0.94)	(0.91)
EPS	+	5.738***	5.758***	5.751***	5.742***
		(3.36)	(3.37)	(3.37)	(3.36)
EPS*DBEAT5	+	4.092***	3.864***	3.464***	4.491***
		(4.62)	(4.56)	(3.54)	(3.43)
EPS*DBEAT5*D_AM_ACCR5	-	-2.197***	-2.050***	-2.214***	-2.011**
		(-2.69)	(-2.68)	(-3.08)	(-2.73)
EPS*DBEAT5*D_RM_SALE5	-	-3.816***			-3.750***
		(-3.47)			(-2.98)
EPS*DBEAT5*D_RM_PROD5	-		-1.385*		-1.008
			(-1.82)		(-1.48)
EPS*DBEAT5*D_RM_DIXP5	-			0.048	-0.330
				(0.05)	(-0.32)
EPS*Growth5	+	$4.796^{**}$	4.756**	$4.788^{**}$	4.6766**
		(2.47)	(2.42)	(2.46)	(2.45)
EPS*Leverage	-	-2.358***	-2.381***	-2.378***	-2.361***
		(-5.81)	(-5.79)	(-5.79)	(-5.82)
EPS*Evar	-	-0.000	-0.000	-0.000	-0.000
		(-1.57)	(-1.57)	(-1.58)	(-1.57)
BVS	+	1.281***	1.279***	1.279***	1.282***
		(3.66)	(3.65)	(3.65)	(3.66)
R-Squared		0.580	0.580	0.579	0.581

PRICE = close price per share at the fiscal year end (Compustat #199). EPS = net income before extraordinary items divided by number of shares outstanding for basic EPS (Compustat #54). DBEAT5 = indicator variable equals 1 if a firm reports a 5-year pattern of increasing positive earnings, and 0 otherwise. D AM ACCR5 = indicator variable equals 1 if earnings without the influence of abnormal total accruals are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. DBEAT5\*D\_AM\_ACCR5 = interaction indicator variable equals 1 if a firm with a fiveyear pattern of increasing earnings presents abnormal total accruals that, if not included in earnings, would fail to report the pattern of increasing earnings, and 0 otherwise. D\_RM\_SALE5 = indicator variable equals 1 if earnings without the influence of sales management are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. DBEAT5\*D RM SALE5 = interaction indicator variable equals 1 if a firm with a five-year pattern of increasing earnings fails to maintain the pattern when the influence of sales management is removed from earnings, and 0 otherwise. D RM PROD5 = indicator variable equals 1 if earnings without the influence of production management are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. DBEAT5\*D RM PROD5 = interaction indicator variable equals 1 if a firm with a five-year pattern of increasing earnings fails to maintain the pattern when the influence of production management is removed from earnings, and 0 otherwise. D RM DIXP5 = indicator variable equals 1 if earnings without the influence of expenses management are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. DBEAT5\*D RM DIXP5 = interaction indicator variable equals 1 if a firm with a five-year pattern of increasing earnings fails to maintain the pattern when the influence of discretionary expense management is removed from earnings, and 0 otherwise. Leverage = sum of short-term debt (Compustat #34) and long-term debt (Compustat #9) divided by market capitalization. Evar = variance of the past 5 years' percentage change in earnings. Growth5 = 5-year compound growth rate of book value of equity. BVS = book value of equity per share.

Table 6
Analysis of Market Rewards to Non-Suspect Beaters and Suspect Firms

N=22,605		1	2	3	4
Variable	Predicted	Coef.	Coef.	Coef.	Coef.
variable	Sign	(t-statistic)	(t-statistic)	(t-statistic)	(t-statistic)
Intercept	?	1.018	1.029	1.043	1.050
		(0.93)	(0.94)	(0.95)	(0.96)
EPS	+	5.757***	5.744***	5.763***	5.755***
		(3.37)	(3.36)	(3.37)	(3.37)
EPS*DBEAT5	+	2.923**	3.011***	2.836***	2.601***
		(4.03)	(4.12)	(3.77)	(2.85)
EPS*DBEAT5*DCFO5	+	3.567***	3.114**	3.570**	3.728**
		(2.61)	(2.44)	(2.66)	(2.73)
EPS*DBEAT5*D_AM_ACCR5	-	-2.050***			
		(-2.66)			
EPS*DBEAT5*D_RM_SALE5	-		-3.298***		
			(-3.16)		
EPS*DBEAT5*D_RM_PROD5	-			-1.389*	
				(-1.83)	
EPS*DBEAT5*D_RM_DIXP5	-				-0.455
		**	**	**	(-0.45)
EPS*Growth5	+	4.738**	4.757**	4.711**	4.731**
		(2.40)	(2.43)	(2.37)	(2.40)
EPS*Leverage	-	-2.371***	-2.366***	-2.385***	-2.381***
		(-5.78)	(-5.84)	(-5.81)	(-5.80)
EPS*Evar	-	-0.000	-0.000	-0.000	-0.00
		(-1.57)	(-1.57)	(-1.58)	(-1.58)
BVS	+	1.280***	1.282***	1.280***	1.280***
		(3.65)	(3.66)	(3.65)	(3.65)
R-Squared		0.580	0.580	0.580	0.580

PRICE = close price per share at the fiscal year end (Compustat #199). EPS = net income before extraordinary items divided by number of shares outstanding for basic EPS (Compustat #54). DBEAT5 = indicator variable equals 1 if a firm reports a 5-year pattern of increasing positive earnings, and 0 otherwise. DCFO5 = indicator variable equals 1 if a firm reports a 5-year pattern of increasing positive cash flows, and 0 otherwise. DBEAT5\*DCFO5 = interaction indicator variable equals 1 if a firm reports five-year patterns both in earnings increases and in cash flows increases, and 0 otherwise. D\_AM\_ACCR5 = indicator variable equals 1 if earnings without the influence of abnormal total accruals are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. DBEAT5\*D AM ACCR5 = interaction indicator variable equals 1 if a firm with a five-year pattern of increasing earnings presents abnormal total accruals that, if not included in earnings, would fail to report the pattern of increasing earnings, and 0 otherwise. D RM SALE5 = indicator variable equals 1 if earnings without the influence of sales management are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. DBEAT5\*D RM SALE5= interaction indicator variable equals 1 if a firm with a five-year pattern of increasing earnings fails to maintain the pattern when the influence of sales management is removed from earnings, and 0 otherwise. D RM PROD5 = indicator variable equals 1 if earnings without the influence of production management are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. DBEAT5\*D RM PROD5 = interaction indicator variable equals 1 if a firm with a five-year pattern of increasing earnings fails to maintain the pattern when the influence of production management is removed from earnings, and 0 otherwise. D RM DIXP5 = indicator variable equals 1 if earnings without the influence of expenses management are less than previous year's actual earnings during any consecutive five years, and 0 otherwise. DBEAT5\*D RM DIXP5 = interaction indicator variable equals 1 if a firm with a fiveyear pattern of increasing earnings fails to maintain the pattern when the influence of discretionary expense management is removed from earnings, and 0 otherwise.

Leverage = sum of short-term debt (Compustat #34) and long-term debt (Compustat #9) divided by market capitalization. Evar = variance of the past 5 years' percentage change in earnings. Growth5 = 5-year compound growth rate of book value of equity. BVS = book value of equity per shares.

Table 7
Analysis of Market Rewards to Firms with a Pattern of Increasing Earnings Not Achieved through Sales Growth

Panel A: D	Panel A: Descriptive Statistics							
Num. of	Num. of	Num. of	Num. of Suspect Firms					
Years	<b>Total Observations</b>	Earnings Beaters	(% over Earnings Beaters)					
5	22605	2088	398(0.19)					
Panel B: Re	egression Results							
N=22,605								
Variable		Predicted	Coef.					
		Sign	(t-statistic)					
Intercept		?	1.058					
			(0.97)					
EPS		+	5.753***					
			(3.37)					
EPS*DBEAT5		+ 3.350***						
			(4.73)					
EPS*DBEAT5*D_NONREV5		-	-2.70***					
	_		(-3.80)					
EPS*Grow	th5	+	4.729**					
			(2.38)					
EPS*Lever	rage	-	-2.383***					
			(-5.81)					
EPS*Evar		-	-0.000					
			(-1.58)					
BVS		+	1.280***					
			(3.66)					
R-Squared			0.580					

PRICE = close price per share at the fiscal year end (Compustat #199). EPS = net income before extraordinary items divided by number of shares outstanding for basic EPS (Compustat #54). DBEAT5 = indicator variable equals 1 if a firm reports a 5-year pattern of increasing positive earnings, and 0 otherwise. DBEAT5\*D\_NONREV5 = interaction indicator variable equals 1 if a firm reports a five-year pattern in earnings increases but not sales growth, and 0 otherwise. Leverage = sum of short-term debt (Compustat #34) and long-term debt (Compustat #9) divided by market capitalization. Evar = variance of the past 5 years' percentage change in earnings. Growth5 = 5-year compound growth rate of book value of equity. BVS = book value of equity per share.

## **Appendix**

Variable Descriptions and Definitions

PRICE Close price – fiscal year end (#199)

Earnings NIBE, net income before extraordinary items (#18)

EPS net income before extraordinary items divided by number of shares

outstanding for basic EPS (#54)

DBEAT5 An indicator variable equals to 1 if a firm reports a 5-year pattern of

increasing positive earnings, and 0 otherwise

CFO Cash flow from operations (#308)

DCFO5 An indicator variable equals 1 if a firm reports a 5-year pattern of

increasing cash flows, and 0 otherwise

DBEAT5\*DCFO5 An interaction indicator variable equals 1 if a firm reports 5-year

patterns both in earnings increases and in cash flows increases, and 0

otherwise

A Total assets (#6)

Total Accruals the difference between net income before extraordinary items (#18) and

cash flows from operations (#308) divided by lagged total assets (#6)

PPE Firm's gross property, plant, and equipment, COMPUSTAT #7
Ab\_TA the level of abnormal total accruals computed using the Jones Model
D AM ACCR5 An indicator variable takes value of 1 if earnings without discretionary

An indicator variable takes value of 1 if earnings without discretionary accruals (abnormal total accruals multiplied by lagged total assets) are less than previous year's actual earnings during any of the five years,

and 0 otherwise

DBEAT5\*D AM ACCR5 An interaction indicator variable takes value of 1 if a firm with a five-

year pattern of increasing earnings presents discretionary total accruals that, if not included in earnings, would fail to report the pattern of

increasing earnings, and 0 otherwise

Sales Net sales (#12)

Ab\_CFO the level of abnormal cash flows from operations computed using

equation (3)

D\_RM\_SALES5 An indicator variable takes value of 1 if earnings without sales

management (abnormal CFO multiplied by lagged total assets) are less than previous year's actual earnings during any of the five years, and 0

otherwise

DBEAT5\*D RM SALE5 An interaction indicator variable takes value of 1 if a firm with a five-

year pattern of increasing earnings fails to maintain the pattern when the influence of sales management is not included in earnings, and 0

otherwise

COG Cost of goods sold (#41)

INVT Inventory (#3)

PROD Production cost = COGS (#41) +  $\Delta$ INVT (#3)

Ab\_PROD the level of abnormal production costs computed using equation (4)
D\_RM\_PROD5 the level of abnormal production costs computed using equation (4)
An indicator variable takes value of 1 if earnings without production

management (abnormal production costs multiplied by total assets) are less than previous year's actual earnings during any of the five years,

and 0 otherwise

DBEAT5\*D\_RM\_PROD5 An interaction indicator variable takes value of 1 if a firm with a five-

year pattern of increasing earnings fails to maintain the pattern when production management is not included in earnings, and 0 otherwise

R&D expenses (#46)

SGA Selling, general and administrative expenses (#189)

Advertising Advertising expenses (#45)

Discretionary Expenses R&D (#46) + SGA (#189) + Advertising (#45)
Ab\_DIXP the level of abnormal discretionary expenses

	by total assets) are less than previous year's actual earnings during any consecutive five years, and 0 otherwise
DBEAT5*D_RM_DIXP5	An interaction indicator variable takes value of 1 if a firm with a five- year pattern of increasing earnings fails to maintain the pattern when the influence of discretionary expense management is not included in earnings, and 0 otherwise
DBEAT5*D_NONREV5	An interaction indicator variable equals 1 if a firm reports a five-year pattern of increasing earnings but not a five-year pattern of increasing
	sales, and 0 otherwise.
Leverage	The sum of short-term debt (#34) and long-term debt (#9) divided by market capitalization(#199*#25)
Evar	Variance of the past five years' percentage change in earnings
	$(NIBE_{i,t} - NIBE_{i,t-1})/abs(NIBE_{i,t-1})$
Growth5	Five-year compound annual growth rate of book value of equity (#60),
	$(BV_t/BV_{t-6})^{1/5}$ -1
BVS	Book value of equity per share (#60/# 54).

All variables are in millions of dollars except number of common shares outstanding for basic EPS (#54), which units are millions.

## **Edicions / Issues:**

95/1	Productividad del trabajo, eficiencia e hipótesis de convergencia en la industria textil-confección europea Jordi López Sintas
95/2	El tamaño de la empresa y la remuneración de los máximos directivos Pedro Ortín Ángel
95/3	Multiple-Sourcing and Specific Investments Miguel A. García-Cestona
96/1	La estructura interna de puestos y salarios en la jerarquía empresarial Pedro Ortín Ángel
96/2	Efficient Privatization Under Incomplete Contracts Miguel A. García-Cestona Vicente Salas-Fumás
96/3	Institutional Imprinting, Global Cultural Models, and Patterns of OrganizationalLearning: Evidence from Firms in the Middle-Range Countries Mauro F. Guillén (The Wharton School, University of Pennsylvania)
96/4	The relationship between firm size and innovation activity: a double decision approach Ester Martínez-Ros (Universitat Autònoma de Barcelona) José M. Labeaga (UNED & Universitat Pompeu Fabra)
96/5	An Approach to Asset-Liability Risk Control Through Asset-Liability Securities Joan Montllor i Serrats María-Antonia Tarrazón Rodón
97/1	Protección de los administradores ante el mercado de capitales: evidencia empírica en España Rafael Crespí i Cladera
97/2	Determinants of Ownership Structure: A Panel Data Approach to the Spanish Case Rafael Crespí i Cladera
97/3	The Spanish Law of Suspension of Payments: An Economic Analysis From Empirical Evidence Esteban van Hemmen Almazor
98/1	Board Turnover and Firm Performance in Spanish Companies Carles Gispert i Pellicer
98/2	Libre competencia frente a regulación en la distribución de medicamentos: teoría y evidencia empírica para el caso español Eva Jansson
98/3	Firm's Current Performance and Innovative Behavior Are the Main Determinants of Salaries in Small-Medium Enterprises

	Jordi López Sintas y Ester Martínez Ros
98/4	On The Determinants of Export Internalization: An Empirical Comparison Between Catalan and Spanish (Non-Catalan) Exporting Firms Alex Rialp i Criado
98/5	Modelo de previsión y análisis del equilibrio financiero en la empresa Antonio Amorós Mestres
99/1	Avaluació dinàmica de la productivitat dels hospitals i la seva descomposició en canvi tecnològic i canvi en eficiència tècnica Magda Solà
99/2	Block Transfers: Implications for the Governance of Spanish Corporations Rafael Crespí, and Carles Gispert
99/3	The Asymmetry of IBEX-35 Returns With TAR Models M.a Dolores Márquez, César Villazón
99/4	Sources and Implications of Asymmetric Competition: An Empirical Study Pilar López Belbeze
99/5	El aprendizaje en los acuerdos de colaboración interempresarial Josep Rialp i Criado
00/1	The Cost of Ownership in the Governance of Interfirm Collaborations Josep Rialp i Criado, i Vicente Salas Fumás
00/2	Reasignación de recursos y resolución de contratos en el sistema concursal español Stefan van Hemmen Alamazor
00/3	A Dynamic Analysis of Intrafirm Diffusion: The ATMs Lucio Fuentelsaz, Jaime Gómez, Yolanda Polo
00/4	La Elección de los Socios: Razones para Cooperar con Centros de Investigación y con Proveedores y Clientes Cristina Bayona, Teresa García, Emilio Huerta
00/5	Inefficient Banks or Inefficient Assets? Emili Tortosa-Ausina
01/1	Collaboration Strategies and Technological Innovation: A Contractual Perspective of the Relationship Between Firms and Technological Centers Alex Rialp, Josep Rialp, Lluís Santamaria
01/2	Modelo para la Identificación de Grupos Estratégicos Basado en el Análisis Envolvente de Datos: Aplicación al Sector Bancario Español Diego Prior, Jordi Surroca
01/3	Seniority-Based Pay: Is It Used As a Motivation Device? Alberto Bayo-Moriones
01/4	Calidad de Servicio en la Enseñanza Universitaria: Desarrollo y Validación de una Escala de Medida.

	Joan-Lluís Capelleras, José M.ª Veciana
01/5	Enfoque estructural vs. recursos y capacidades: un estudio empírico de los factores clave de éxito de las agencias de viajes en España.  Fabiola López-Marín, José M.ª Veciana
01/6	Opción de Responsabilidad Limitada y Opción de Abandonar: Una Integración para el Análisis del Coste de Capita.  Neus Orgaz
01/7	Un Modelo de Predicción de la Insolvencia Empresarial Aplicado al Sector Textil y Confección de Barcelona (1994-1997). Antonio Somoza López
01/8	La Gestión del Conocimiento en Pequeñas Empresas de Tecnología de la Información: Una Investigación Exploratoria. Laura E. Zapata Cantú
01/9	Marco Institucional Formal de Creación de Empresas en Catalunya: Oferta y Demanda de Servicios de Apoyo David Urbano y José María Veciana.
02/1	Access as a Motivational Device: Implications for Human Resource Management. Pablo Arocena, Mikel Villanueva
02/2	Efficiency and Quality in Local Government. The Case of Spanish Local Authorities M.T. Balaguer, D. Prior, J.M. Vela
02/3	Single Period Markowitz Portfolio Selection, Performance Gauging and Duality: A variation on Luenberger's Shortage Function Walter Briec, Kristiaan Kerstens, Jean Baptiste Lesourd
02/4	Innovación tecnológica y resultado exportador: un análisis empírico aplicado al sector textil-confección español Rossano Eusebio, Àlex Rialp Criado
02/5	Caracterización de las empresas que colaboran con centros tecnológicos Lluís Santamaria, Miguel Ángel García Cestona, Josep Rialp
02/6	Restricción de crédito bancario en economías emergentes: el caso de la PYME en México Esteban van Hemmen Almazor
02/7	La revelación de información obligatoria y voluntaria (activos intangibles) en las entidades de crédito. Factores determinantes.  Gonzalo Rodríguez Pérez
02/8	Measuring Sustained Superior Performance at the Firm Level Emili Grifell - Tatjé, Pilar Marquès - Gou
02/9	Governance Mechanisms in Spanish Financial Intermediaries Rafel Crespi, Miguel A. García-Cestona, Vicente Salas
02/10	Endeudamiento y ciclos políticos presupuestarios: el caso de los ayuntamientos

	catalanes Pedro Escudero Fernández, Diego Prior Jiménez
02/11	The phenomenon of international new ventures, global start-ups, and born-globals:what do we know after a decade (1993-2002) of exhaustive scientific inquiry? Àlex Rialp-Criado, Josep Rialp-Criado, Gary A. Knight
03/1	A methodology to measure shareholder value orientation and shareholder value creation aimed at providing a research basis to investigate the link between both magnitudes  Stephan Hecking
03/2	Assessing the structural change of strategic mobility. Determinants under hypercompetitive environments José Ángel Zúñiga Vicente, José David Vicente Lorente
03/3	Internal promotion versus external recruitment: evidence in industrial plants Alberto Bayo-Moriones, Pedro Ortín-Ángel
03/4	El empresario digital como determinante del éxito de las empresas puramente digitales: un estudio empírico Christian Serarols, José M.ª Veciana
03/5	La solvencia financiera del asegurador de vida y su relación con el coste de capital Jordi Celma Sanz
03/6	Proceso del desarrollo exportador de las empresas industriales españolas que participan en un consorcio de exportación: un estudio de caso Piedad Cristina Martínez Carazo
03/7	Utilidad de una Medida de la Eficiencia en la Generación de Ventas para la Predicción del Resultado María Cristina Abad Navarro
03/8	Evaluación de fondos de inversión garantizados por medio de portfolio insurance Sílvia Bou Ysàs
03/9	Aplicación del DEA en el Análisis de Beneficios en un Sistema Integrado Verticalmente Hacia Adelante Héctor Ruiz Soria
04/1	Regulación de la Distribución Eléctrica en España: Análisis Económico de una Década, 1987-1997 Leticia Blázquez Gómez; Emili Grifell-Tatjé
04/2	The Barcelonnettes: an Example of Network-Entrepreneurs in XIX Century Mexico.  An Explanation Based on a Theory of Bounded Rational Choice with Social Embeddedness.  Gonzalo Castañeda
04/3	Estructura de propiedad en las grandes sociedades anónimas por acciones. Evidencia empírica española en el contexto internacional Rabel Crespí; Eva Jansson

05/1	TEDGAL C C CC
05/1	IFRS Adoption in Europe: The Case of Germany.
	Soledad Moya, Jordi Perramon, Anselm Constans
05/2	Efficiency and environmental regulation: a 'complex situation'
00/2	Andrés J. Picazo-Tadeo, Diego Prior
05/3	Financial Development, Labor and Market Regulations and Growth
	Raquel Fonseca, Natalia Utrero
06/1	Entrepreneurship, Management Services and Economic Growth
	Vicente Salas Fumás, J. Javier Sánchez Asín
06/2	Triple Bottom Line: A business metaphor for a social construct
	Darrel Brown, Jesse Dillard, R. Scott Marshall
06/3	El Riesgo y las Estrategias en la Evaluación de los Fondos de Inversión de Renta
	Variable
	Sílvia Bou
06/4	Corporate Governance in Banking: The Role of Board of Directors
	Pablo de Andrés Alonso, Eleuterio Vallelado González
06/5	The Effect of Relationship Lending on Firm Performance
	Judit Montoriol Garriga
0.7/7	
06/6	Demand Elasticity and Market Power in the Spanish Electricity Market
	Aitor Ciarreta, María Paz Espinosa
0.6/7	
06/7	Testing the Entrepreneurial Intention Model on a Two-Country Sample
	Francisco Liñán, Yi-Wen Chen
07/1	Took a logical transpolines for a surrenting question in Catalonia, the ease of the
07/1	Technological trampolines for new venture creation in Catalonia: the case of the University of Girona
	Andrea Bikfalvi, Christian Serarols, David Urbano, Yancy Vaillant
	Andrea Dikiarvi, Christian Scrarois, David Croano, Tancy Vaniant
07/2	Public Enterprise Reforms and Efficiency in Regulated Environments: the Case of
<u>-</u>	the Postal Sector
	Juan Carlos Morales Piñero, Joaquim Vergés Jaime
	1
07/3	The Impact of Prevention Measures and Organisational Factors on Occupational
	Injuries
	Pablo Arocena, Imanol Núñez, Mikel Villanueva
07/4	El impacto de la gestión activa en la performance de los fondos de inversión de
	renta fija
	Sílvia Bou Ysàs
07/5	Organisational status and efficiency: The case of the Spanish SOE "Paradores"
	Magda Cayón, Joaquim Vergés
07/6	Longitudinal Analysis of Enterpreneurship and competitiveness dynamics in Latin

	America
	José Ernesto Amorós, Óscar Cristi
08/1	Earnings Management and cultural values
	Kurt Desender, Christian Castro, Sergio Escamilla
08/2	Why do convertible issuers simultaneously repurchase stock? An arbitrage-based explanation
	Marie Dutordoir, Patrick Verwijmeren
08/3	Entrepreneurial intention, cognitive social capital and culture: empirical analysis for Spain and Taiwan
	Francisco Liñán, Francisco Santos, José L. Roldán
08/4	From creative ideas to new emerging ventures: the process of identification and exploitation among finnish design entrepreneurs  Henrik Tötterman
	Henrik Totterman
08/5	Desempeño de la Política Comercial Pública en España
00/2	Manuel Sánchez, Ignacio Cruz, David Jiménez
00/6	
08/6	Gender Effects on Performance in Bulgarian Private Enterprises  Desislava Yordanova
	Desisiava Yordanova
08/7	Entorno e iniciativa emprendedora: una perspectiva internacional
	Claudia Álvarez, David Urbano
09/1	Narrating Urban Entrepreneurship: A Matter of Imagineering?
	Chris Steyaert, Timon Beyes
09/2	Organizational Configurations of Strategic Choices and Strategic Management
07/2	Accounting
	Simon Cadez, Chris Guilding
09/3	Agency Cost of Government Ownership: A study of Voluntary Audit Comitte Formation in China
	David Hillier, Charlie X. Cai, Gaoliang Tian, Qinghua Wu
09/4	Public Policy for Entropyon curshin and Innovation, Impact in Managed and
09/4	Public Policy for Entrepreneurship and Innovation: Impact in Managed and Entrepreneurial Economies
	Karen Murdock
09/5	Glocalization as a Generic Entrepreneurial Strategy
	Bengt Johanisson
09/6	Associng Advertising Efficiency: Does the Internet Play a Role?
09/0	Assesing Advertising Efficiency: Does the Internet Play a Role?  Albena Pergelova, Diego Prior, Josep Rialp
	7 Hoena Tergelova, Diego Thor, Josep Klarp
09/7	Start-up Conditions and the Performance of Women – and Men- Controlled
	Businesses in Manufacturating Industries
	Otilia Driga, Diego Prior

10/1	Devolution Dynamics of Spanish Local Government
	Maria Teresa Balaguer-Coll, Diego Prior, Emili Tortosa-Ausina
10/2	
10/2	Los derivados financieros como herramienta para evaluar la reforma laboral: una aproximación binomial
	Sílvia Bou, Albert Hernández, Carlota Linares
	Silvia Bou, Albert Hernandez, Cariota Emarcs
10/3	Environmental Factors And Social Entrepreneurship
	Elisabeth Ferri, David Urbano
10/4	Accounting Conservatism and Firm Investment Efficiency
	Beatriz García, Juan Manuel García, Fernando Penalva
10/5	The Complementarity Between Segment Disclosure and Earnings Quality, and its
	Effect on Cost of Capital  Belén Blanco, Juan M. García, Josep A. Tribó
	Beien Blanco, Juan W. Garcia, Josep A. 11100
10/6	Revisiting the Size-R&D Productivity Relation: Introducing the Mediating Role of
	Decision-Making Style on the Scale and Quality of Innovative Output
	José Lejarraga, Ester Martínez
10/7	
10/7	Nuevos y viejos criterios de rentabilidad que concuerdan con el criterio del Valor
	Actual Neto Emilio Padilla, Joan Pascual
	Ellillo Fadilla, Joan Fascual
10/8	A cognitive attempt to understanding female entrepreneurial potential: the role of
10,0	social norms and culture
	Francisco Liñán, Muhammad A. Roomi, Francisco J. Santos
11/1	Behavioral Aspects of Investment Fund's Markets: Are Good Managers Lucky or
	Skilled? Sílvia Bou, Magda Cayón
	Silvia Bou, Magua Cayon
11/2	
	Place Marketing Performance: Benchmarking European Cities as Business Destinations.
	Albena Pergelova
	Thousa Tergerova
11/3	Portfolio Selection with Skewness: A Comparison of Methods and a Generalized
	Two Fund Separation Result
	Walter Briec, Kristiaan Kerstens, Ignace Van de Woestyne
11/4	Can proprietional commitment by a services 1 1 1 1 1 1 1 1 1 1
	Can organizational commitment be experienced by individuals pursuing contemporary career paths?
	Mihaela Enache, Jose M. Sallan, Pep Simo and Vicenc Fernandez
	namem Limine, sose m. Samin, i ep suno una vicena i entanuez
11/5	Social Capital and the Equilibrium Number of Enterpreneurs.
	Vicente Salas-Fumás, J.Javier Sanchez-Asin
11/6	Determinants of Acquisition Completion: A Relational Perspective.
	Ruth V. Aguilera, John C. Dencker
11/7	SME's Environmental CSR Investment: Evaluation, Decision and Implication.
	F. Merlinda

12/1	Debt Enforcement and Relational Contracting.
	Martin Brown, Marta Serra-García
1.7/2	
12/2	Stockholder Conflicts and Dividends
	Janis Berzins, Øyvind Bøhren, Bogdan Stacescu
12/3	The "Death of Environmentalism" Debates: Forging Links Between SEA and Civil Society.
	Judy Brown, Jesse Dillard
12/4	Evaluación y rediseño de la red pública educativa. Un análisis centralizado.  Laura López, Diego Prior
12/5	Organizational resources and intrapreneurial activities: A cross-country study.
	Andreu Turró, David Urbano
12/6	The Hidden Costs of Hidden Debt.
	Johan Almenberg and Arteshes Karapetyan
12/7	The Effects of Walking while Working on Productivity and Health: A Field Experiment.
	Avner Ben-Ner, Darla Flint Paulson, Gabriel Koepp, James Levine
12/8	How does Ownership Structure influence Bank Risk? Analyzing the Role of Managerial Incentives.
	Mónica López-Puertas Lamy
13/1	Contains the Misses Associated and the Misses Lorentz and Justicians
	Studying the Micro-Angels Approach to Micro-Investment decisions.  Glòria Estapé Dubreuil, Arvind Ashta, Jean-Pierre Hédou
	Gioria Estape Dubleun, Ai vina Asina, sean-i ierre ricuoa
13/2	Multimarket Contact Externalities: The Effect of Rival's Multimarket Contacts on Focal Firm Performance.
	Jaime Gómez, Raquel Orcos, Sergio Palomas
12/2	M. L. D. W. C. L. E. D. C. L. El. D.
13/3	Market Rewards to Patterns of Increasing Earning: Do Cash Flow Patterns, Accruals Manipulation and Real Activities Manipulation Matter?
	Su-Ping Liu, Juan Manuel García Lara